Ulipristal acetate versus levonorgestrel-releasing intrauterine system for heavy menstrual bleeding: the UCON randomised controlled trial and mechanism of action study

Lucy HR Whitaker,¹ Lee J Middleton,² Lee Priest,² Smita Odedra,² Versha Cheed,² Elaine P Nicholls,³ Alistair RW Williams,⁴ Neil Roberts,¹ Clive E Stubbs,² Konstantios Tryposkiadis,² Hannah Bensoussane,² Rohan Chodankar,¹ Alison A Murray,¹ Moira Nicol,¹ Aleksandra O Tsolova,¹ Kaiming Yin,¹ Marcos Cruz,⁵ Hui Wei Leow,¹ Lucy E Kershaw,⁶ Suzanne L McLenachan,⁷ Graham McKillop,⁷ Jane Walker,⁸ Scott I Semple,⁶ T Justin Clark,⁹ Mary Ann Lumsden,¹⁰ Dharani K Hapangama,¹¹ Lucky Saraswat,¹² Siladitya Bhattacharya,¹² Paul Smith,¹³ Jane Daniels¹⁴ and Hilary OD Critchley^{1*}

¹MRC Centre for Reproductive Health, University of Edinburgh, Edinburgh, UK ²Birmingham Clinical Trials Unit, University of Birmingham, Birmingham, UK ³Adcal H.R. Consultancy

- ⁴Division of Pathology, University of Edinburgh, Edinburgh, UK
- ⁵Department of Mathematics, Statistics and Computer Science, University of Cantabria, Spain
- ⁶Centre for Cardiovascular Science, University of Edinburgh, Edinburgh, UK
- ⁷Department of Clinical Radiology, Royal Infirmary of Edinburgh, Edinburgh, UK
- ⁸Simpson Centre for Reproductive Health, Royal Infirmary, Edinburgh, UK
- ⁹Birmingham Women's and Children's Hospital, Birmingham, UK
- ¹⁰Reproductive and Maternal Medicine, University of Glasgow, Glasgow, UK
- ¹¹Department of Women's and Children's Health, Institute of Life Course and Medical Sciences, University of Liverpool, Liverpool, UK
- ¹²University of Aberdeen, Aberdeen, UK
- ¹³University of Birmingham, Birmingham, UK
- ¹⁴Nottingham Clinical Trials Unit, University of Nottingham, Nottingham UK

*Corresponding author hilary.critchley@ed.ac.uk

Disclosure of interests of authors

Full disclosure of interests: Completed ICMJE forms for all authors, including all related interests, are available in the toolkit on the NIHR Journals Library report publication page at https://doi.org/10.3310/FGLQ1687.

Primary conflicts of interest: L.H.R. Whitaker receives grant funding from NIHR (NIHR129801). A.R.W. Williams has received consultancy fees (divided with the University of Edinburgh) from Bayer AG. R.R. Chodankar has been supported as a Clinical Research Fellow by Bayer AG between April 2018 to February 2021. T.J. Clark has received honoraria from Bayer AG for attending one advisory board meeting (2015) and hands-on training of three clinicians in Essure sterilisation (made by Bayer; 2016) and from Gedeon Richter as part of the Orbis educational programme in Women's Health (sponsored by GR since 2016) and for travel and accommodation expenses to attend FIGO in Rome (2012). He has been President of the British Society for Gynaecological Endoscopy (2019-22). He has been a member of the HTA Prioritisation Committee B (In hospital care; 2017-22) and HTA Prioritisation Committee C (mental health, women and children's health; 2017–22). He has current research funding as the lead, or as a co-applicant, from the NIHR HTA Programme for the following research projects: NIHR130310, NIHR128991, NIHR127280, NIHR 129801. M.A. Lumsden has advised Gedeon Richter (2018). D.K. Hapangama receives grant funding from Wellbeing of Women (RG213), MRC (MR/V007238/1) and Northwest Cancer Research (RDG2021.18) as well as honoraria from Canadian Society of Fertility and Andrology. L. Saraswat reports grants from NIHR for conduct of trial (NIHR127280; 2019) and receiving honorarium from Gideon-Richter for attending one advisory board meeting (2022). S. Bhattacharya receives royalties from Cambridge University Press and payment (to University of Aberdeen) for speaking at 11th Singapore International Congress on Obstetrics and Gynaecology, and invited lectures to Merck and Ferring. He is a board member of NHS Grampian for which the University of Aberdeen receives payment. He receives an honorarium from Oxford University Press for his role as Editor in Chief (Human Reproduction Open) is special Senior Editor, Cochrane Gynaecology and Infertility (no honorarium). P. Smith receives grant funding from an NIHR Fellowship (PDF-2015-08-099). J. Daniels is a member of NIHR CTU Standing Advisory Committee. H.O.D. Critchley has received grant funding (paid to institution) from the Biotechnology and Biological Sciences Research Council, grants from the Medical Research Council/NIHR to support salaries for research staff and study consumables and a research collaboration grant from Bayer AG, Berlin, with salaries for research staff and study consumables. She has personal receipt of royalties from 'Up-To-Date' for an article on abnormal uterine bleeding. She has received consulting fees paid to institution from Bayer AG (Consultancy and Scientific Advisory Board advice; no personal remuneration received), Gedeon Richter (Consultancy advice; no personal remuneration received) and Myovant Sciences GmbH (Consultancy and Scientific Advisory Board advice; no personal remuneration received). She has received speaker fees from Vifor Pharma UK Ltd (with no personal remuneration received) and travel expenses for attendance(s) at SAB (Scientific Advisory Board). She is Chair (from 2021-23) of the Committee for Menstrual Disorders and Related Health Impacts of the International Federation of Gynecology and Obstetrics (no payment received).

Funding: This project was funded by the National Institute for Health and Care Research (NIHR) Efficacy and Mechanism Evaluation Programme and will be published in full in *Efficacy and Mechanism Evaluation* Vol. 10, No. 8. See the NIHR Journals Library website for further project information.

Published October 2023 DOI: 10.3310/FGLQ1687

Plain language summary

Ulipristal acetate versus levonorgestrel-releasing intrauterine system for heavy menstrual bleeding: the UCON randomised controlled trial and mechanism of action study

Efficacy and Mechanism Evaluation 2023; Vol. 10: No. 8 DOI: 10.3310/FGLQ1687

NIHR Journals Library www.journalslibrary.nihr.ac.uk

Plain language summary

What is the problem?

Heavy menstrual bleeding is a common condition that affects the lives of many women. A hormonereleasing coil, fitted inside the womb, is effective in making periods lighter but can make them less regular. A medicine called ulipristal acetate or UPA, taken as a pill, has been shown to reduce rapidly menstrual bleeding in women with large, non-cancerous tumours in the womb, known as fibroids. It was not known whether UPA is effective in women who have heavy periods but do not have fibroids of any significant size.

What did we plan to do?

To find out which treatment was better at controlling heavy periods, 236 women were enrolled in a clinical trial where they received either the hormone coil or UPA. The choice of treatment was made at random by a computer rather than the wishes of researchers or patients, to ensure a fair comparison. Participants completed questionnaires about their symptoms and life quality at intervals up to 1 year.

Twice during the trial, medicines regulatory authorities raised safety concerns about UPA causing liver problems. This resulted in the introduction of regular blood tests. The second time, recruitment to the trial stopped early.

What did we find?

Both treatments improved the symptoms of heavy menstrual bleeding in the majority of women. We found no evidence that UPA was better overall after 1 year of treatment, compared with the hormone coil, although fewer women on UPA continued to have periods. Laboratory studies on samples taken from the lining of the womb showed temporary changes due to UPA, which disappeared after treatment stopped.

What does this mean?

Both treatments improve the symptoms of menstrual bleeding and general wellbeing. Because of safety concerns UPA is not available for all women with heavy menstrual bleeding and new, safer medical treatments are needed.

Efficacy and Mechanism Evaluation

ISSN 2050-4365 (Print)

ISSN 2050-4373 (Online)

Efficacy and Mechanism Evaluation (EME) was launched in 2014 and is indexed by Europe PMC, DOAJ, Ulrichsweb[™] (ProQuest LLC, Ann Arbor, MI, USA) and NCBI Bookshelf.

This journal is a member of and subscribes to the principles of the Committee on Publication Ethics (COPE) (www.publicationethics.org/).

Editorial contact: journals.library@nihr.ac.uk

The full EME archive is freely available to view online at www.journalslibrary.nihr.ac.uk/eme.

Criteria for inclusion in the Efficacy and Mechanism Evaluation journal

Reports are published in *Efficacy and Mechanism Evaluation* (EME) if (1) they have resulted from work for the EME programme, and (2) they are of a sufficiently high scientific quality as assessed by the reviewers and editors.

EME programme

The Efficacy and Mechanism Evaluation (EME) programme funds ambitious studies evaluating interventions that have the potential to make a step-change in the promotion of health, treatment of disease and improvement of rehabilitation or long-term care. Within these studies, EME supports research to improve the understanding of the mechanisms of both diseases and treatments.

The programme supports translational research into a wide range of new or repurposed interventions. These may include diagnostic or prognostic tests and decision-making tools, therapeutics or psychological treatments, medical devices, and public health initiatives delivered in the NHS.

The EME programme supports clinical trials and studies with other robust designs, which test the efficacy of interventions, and which may use clinical or well-validated surrogate outcomes. It only supports studies in humans and where there is adequate proof of concept. The programme encourages hypothesis-driven mechanistic studies, integrated within the efficacy study, that explore the mechanisms of action of the intervention or the disease, the cause of differing responses, or improve the understanding of adverse effects. It funds similar mechanistic studies linked to studies funded by any NIHR programme.

The EME programme is funded by the Medical Research Council (MRC) and the National Institute for Health and Care Research (NIHR), with contributions from the Chief Scientist Office (CSO) in Scotland and National Institute for Social Care and Health Research (NISCHR) in Wales and the Health and Social Care Research and Development (HSC R&D), Public Health Agency in Northern Ireland.

This report

The research reported in this issue of the journal was funded by the EME programme as project number EME 12/206/52. The contractual start date was in October 2014. The final report began editorial review in October 2021 and was accepted for publication in August 2022. The authors have been wholly responsible for all data collection, analysis and interpretation, and for writing up their work. The EME editors and production house have tried to ensure the accuracy of the authors' report and would like to thank the reviewers for their constructive comments on the final report document. However, they do not accept liability for damages or losses arising from material published in this report.

This report presents independent research. The views and opinions expressed by authors in this publication are those of the authors and do not necessarily reflect those of the NHS, the NIHR, the MRC, the EME programme or the Department of Health and Social Care. If there are verbatim quotations included in this publication the views and opinions expressed by the interviewees are those of the interviewees and do not necessarily reflect those of the authors, those of the NHS, the NIHR, the EME programme or the Department of Health and Social Care.

Copyright © 2023 Whitaker *et al.* This work was produced by Whitaker *et al.* under the terms of a commissioning contract issued by the Secretary of State for Health and Social Care. This is an Open Access publication distributed under the terms of the Creative Commons Attribution CC BY 4.0 licence, which permits unrestricted use, distribution, reproduction and adaptation in any medium and for any purpose provided that it is properly attributed. See: https://creativecommons.org/licenses/by/4.0/. For attribution the title, original author(s), the publication source – NIHR Journals Library, and the DOI of the publication must be cited.

Published by the NIHR Journals Library (www.journalslibrary.nihr.ac.uk), produced by Newgen Digitalworks Pvt Ltd, Chennai, India (www.newgen.co).

NIHR Journals Library Editor-in-Chief

Dr Cat Chatfield Director of Health Services Research UK

NIHR Journals Library Editors

Professor Andrée Le May Chair of NIHR Journals Library Editorial Group (HSDR, PGfAR, PHR journals) and Editorin-Chief of HSDR, PGfAR, PHR journals

Dr Peter Davidson Interim Chair of HTA and EME Editorial Board, Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Matthias Beck Professor of Management, Cork University Business School, Department of Management and Marketing, University College Cork, Ireland

Dr Tessa Crilly Director, Crystal Blue Consulting Ltd, UK

Dr Eugenia Cronin Consultant in Public Health, Delta Public Health Consulting Ltd, UK

Ms Tara Lamont Senior Adviser, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Catriona McDaid Reader in Trials, Department of Health Sciences, University of York, UK

Professor William McGuire Professor of Child Health, Hull York Medical School, University of York, UK

Professor Geoffrey Meads Emeritus Professor of Wellbeing Research, University of Winchester, UK

Professor James Raftery Professor of Health Technology Assessment, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Dr Rob Riemsma Consultant Advisor, School of Healthcare Enterprise and Innovation, University of Southampton, UK

Professor Helen Roberts Professor of Child Health Research, Child and Adolescent Mental Health, Palliative Care and Paediatrics Unit, Population Policy and Practice Programme, UCL Great Ormond Street Institute of Child Health, London, UK

Professor Jonathan Ross Professor of Sexual Health and HIV, University Hospital Birmingham, UK

Professor Helen Snooks Professor of Health Services Research, Institute of Life Science, College of Medicine, Swansea University, UK

Please visit the website for a list of editors: www.journalslibrary.nihr.ac.uk/about/editors

Editorial contact: journals.library@nihr.ac.uk